Deborah Saltman, MD, PhD
Editorial Director
BMC Geriatrics

Dear Dr. Saltman,

We thank you for accepting our manuscript, "Predictors of nursing home admission: a Meta-Analysis in the U.S." (1188633327125364) for publication in BMC Geriatrics. Below we outline the remaining formatting changes:

1. References 34-39 are included in Table 2 and Table 3. These reference citations, along with all others in Tables 2 and 3, are now in square brackets as requested. The superscripted text in Figures 2 and 3 was deleted, as this text duplicates the information presented in Tables 2 and 3.

We thank you for considering our work!

Sincerely,

Joseph E. Gaugler, Ph.D.
Assistant Professor
Center on Aging, Center for Gerontological Nursing, School of Nursing
The University of Minnesota
6-150 Weaver-Densford Hall, 1331
308 Harvard Street S.E.
Minneapolis, MN 55455
Email: gaug0015@umn.edu
Telephone: 612-626-2485
Fax: 612-626-2359